Author’s response to reviews

Title: Age-adjusted high-dose chemotherapy and autologous stem cell transplant in elderly and fit primary CNS lymphoma patients

Authors:
Elisabeth Schorb (elisabeth.schorb@uniklinik-freiburg.de)
Juergen Finke (juergen.finke@uniklinik-freiburg.de)
Gabriele Ihorst (gabriele.ihorst@uniklinik-freiburg.de)
Benjamin Kasenda (b.kasenda@klinikum-stuttgart.de)
Heidi Fricker (heidi.fricker@uniklinik-freiburg.de)
Gerald Illerhaus (g.illerhaus@klinikum-stuttgart.de)

Version: 1 Date: 08 Mar 2019

Author’s response to reviews:

Dear Editor,

Please find enclosed our revised manuscript entitled "Age-adjusted high-dose chemotherapy and autologous stem cell transplant in elderly and fit primary CNS lymphoma patients" for review for publication in BMC Cancer.

The required corrections have been carried out as stated below:

1. The text has been modified and overlap to the mentioned publication has been minimised, especially with regard to the Discussion section.

2. 'Methods/Design' has been renamed to 'Methods'.

3. The ethics approval number has been included in the ethics declaration.
4. The TRN has been included in the Editorial Manager Abstract.

5. The section "Supplementary files" containing the mentioned information has been added after the figure legend.

6. We have ensured that all additional files are explicitly referred to in the main text.

7. A single clean copy of the submission has been provided with tables/figures and additional files as separate files.

We are looking forward to hearing from you. Please do not hesitate to contact us in case of any additional questions.

For the authors, sincerely your

Elisabeth Schorb